Author's response to reviews

Title: Opioids for chronic non-cancer pain: a protocol for a systematic review of randomized controlled trials

Authors:

Jason W Busse (j.busse@rogers.com)
Stefan Schandelmaier (s.schandelmaier@gmail.com)
Mostafa Kamaleldin (dr.m.kamal.2@gmail.com)
Sandy Hsu (milkywaysandy@gmail.com)
John J Riva (rivaj@mcmaster.ca)
Per O Vandvik (per.vandvik@gmail.com)
Ludwig Tsoi (tch654@gmail.com)
Tommy Lam (tommylam@yahoo.com)
Shanil Ebrahim (shanil.ebrahim@utoronto.ca)
Bradley Johnston (bradley.johnston@sickkids.ca)
Lori Oliveri (lori.olivieri@gmail.com)
Luis Montoya (montoya.research@gmail.com)
Regina Kunz (Regina.Kunz@usb.ch)
Anna Malandrino (malandrino@mje.utoronto.ca)
Neera Bhatnagar (bhatnag@mcmaster.ca)
Charlene Soobiah (SoobiahC@smh.ca)
Anthony Wong (anthonycpwong@yahoo.com.hk)
Norman Buckley (buckleyn@mcmaster.ca)
Daniel Sessler (ds@or.org)
Gordon H Guyatt (guyatt@mcmaster.ca)
Sohail M Mulla (sohail.mulla@gmail.com)
Luciane C Lopes (luslopes@terra.com.br)

Version: 3 Date: 6 August 2013

Author's response to reviews:

As requested, we have expanded our Authors' Contributions section to include all listed authors as follows:

"JWB and GHG conceived the study design. NB designed the database-specific literature search strategies. AM and SH will retrieve all potentially eligible studies. JWB, SS, MK, SH, JJR, POV, LT, TL, SE, BJ, LO, LM, RK, SMM, LCL and AW will screen potentially eligible studies and abstract data from eligible trials. CS will oversee knowledge translation. LO, NB and DS provided oversight to ensure that our planned subgroup analyses and statistical pooling of effect sizes across patient populations, interventions, controls and outcome measures were clinically sensible. JWB and SS completed the first draft of the manuscript. All authors reviewed several drafts of the manuscript and approved the final version."

Thanks for considering,